

Human T-Lymphotropic Virus type 1c subtype proviral loads, chronic lung disease and survival in a prospective cohort of Indigenous Australians

Lloyd Einsiedel, Hai Pham, Kim Wilson, Rebecca Walley, Jocelyn Turpin,

Charles Bangham, Antoine Gessain, Richard J. Woodman

▶ To cite this version:

Lloyd Einsiedel, Hai Pham, Kim Wilson, Rebecca Walley, Jocelyn Turpin, et al.. Human T-Lymphotropic Virus type 1c subtype proviral loads, chronic lung disease and survival in a prospective cohort of Indigenous Australians. PLoS Neglected Tropical Diseases, 2018, 12 (3), pp.e0006281. 10.1371/journal.pntd.0006281. hal-03183807

HAL Id: hal-03183807 https://hal.inrae.fr/hal-03183807

Submitted on 28 Mar 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Einsiedel L, Pham H, Wilson K, Walley R, Turpin J, Bangham C, et al. (2018) Human T-Lymphotropic Virus type 1c subtype proviral loads, chronic lung disease and survival in a prospective cohort of Indigenous Australians. PLoS Negl Trop Dis 12(3): e0006281. https://doi.org/10.1371/ journal.pntd.0006281

Editor: Eric Mossel, Center for Disease Control and Prevention, UNITED STATES

Received: September 20, 2017

Accepted: January 29, 2018

Published: March 12, 2018

Copyright: © 2018 Einsiedel et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data relates to Indigenous Australians and cannot be accessed without appropriate ethics approval from the Central Australian Human Research Ethics Committee for researchers who meet the criteria for access to confidential data (cahrec@flinders. edu.au).

Funding: This study received funding from the National Health and Medical Research Council of Australia (NHMRC project grant 1012945) and the

RESEARCH ARTICLE

Human T-Lymphotropic Virus type 1c subtype proviral loads, chronic lung disease and survival in a prospective cohort of Indigenous Australians

Lloyd Einsiedel¹*, Hai Pham¹, Kim Wilson², Rebecca Walley³, Jocelyn Turpin⁴, Charles Bangham⁴, Antoine Gessain⁵, Richard J. Woodman⁶

1 Aboriginal Health Domain, Baker Heart and Diabetes Institute central Australia, Alice Springs Hospital, Alice Springs, Australia, 2 National Serology Reference Laboratory, Melbourne, Australia, 3 Flinders University/Northern Territory Rural Clinical School, Alice Springs Hospital, Alice Springs, Australia, 4 Section of Virology, Division of Infectious Diseases, Department of Medicine, Imperial College London, Norfolk Place, London, United Kingdom, 5 Institut Pasteur, Unité d'Epidémiologie et Physiopathologie des Virus Oncogènes, Département de Virologie, Paris, France, CNRS UMR 3569, 6 Flinders Centre for Epidemiology and Biostatistics, Flinders University, Adelaide, Australia

* lloyd.einsiedel@nt.gov.au

Abstract

Background

The Human T-Lymphotropic Virus type 1c subtype (HTLV-1c) is highly endemic to central Australia where the most frequent complication of HTLV-1 infection in Indigenous Australians is bronchiectasis. We carried out a prospective study to quantify the prognosis of HTLV-1c infection and chronic lung disease and the risk of death according to the HTLV-1c proviral load (pVL).

Methodology/Principal findings

840 Indigenous adults (discharge diagnosis of bronchiectasis, 154) were recruited to a hospital-based prospective cohort. Baseline HTLV-1c pVL were determined and the results of chest computed tomography and clinical details reviewed. The odds of an association between HTLV-1 infection and bronchiectasis or bronchitis/bronchiolitis were calculated, and the impact of HTLV-1c pVL on the risk of death was measured.

Radiologically defined bronchiectasis and bronchitis/bronchiolitis were significantly more common among HTLV-1-infected subjects (adjusted odds ratio = 2.9; 95% CI, 2.0, 4.3). Median HTLV-1c pVL for subjects with airways inflammation was 16-fold higher than that of asymptomatic subjects. There were 151 deaths during 2,140 person-years of follow-up (maximum follow-up 8.13 years). Mortality rates were higher among subjects with HTLV-1c pVL \geq 1000 copies per 10⁵ peripheral blood leukocytes (log-rank χ^2 (2df) = 6.63, p = 0.036) compared to those with lower HTLV-1c pVL or uninfected subjects. Excess mortality was largely due to bronchiectasis-related deaths (adjusted HR 4.31; 95% CI, 1.78, 10.42 versus uninfected).

Northern Territory Rural Clinical School, an initiative of the Australian Department of Health and Ageing. CRMB is a Wellcome Trust Senior Investigator. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors declare that no competing interests exist.

Conclusion/Significance

Higher HTLV-1c pVL was strongly associated with radiologically defined airways inflammation and with death due to complications of bronchiectasis. An increased risk of death due to an HTLV-1 associated inflammatory disease has not been demonstrated previously. Our findings indicate that mortality associated with HTLV-1c infection may be higher than has been previously appreciated. Further prospective studies are needed to determine whether these results can be generalized to other HTLV-1 endemic areas.

Author summary

The Human T-Lymphotropic Virus type 1 (HTLV-1) infects up to 20 million people worldwide who predominantly reside in resource-limited areas. The virus is associated with a haematological malignancy (adult T-cell leukaemia/lymphoma, ATL), and inflammatory diseases involving organ systems including the spinal cord, eyes and lungs. Determining the outcomes of infection in most HTLV-1 endemic areas is extremely difficult; however, the virus is highly endemic to central Australia where the Indigenous population has access to sophisticated medical facilities. We prospectively followed a large hospitalbased cohort of Indigenous Australian adults that was well characterized with regard to base-line comorbid conditions, HTLV-1 serostatus and HTLV-1 proviral load (pVL). A higher baseline HTLV-1 pVL was strongly associated with an increased risk of airway inflammation (bronchitis/bronchiolitis and bronchiectasis) and death, which most often resulted from complications of bronchiectasis. Increased mortality due to an HTLV-1associated inflammatory condition has not been demonstrated previously. The morbidity and mortality associated with HTLV-1 infection may therefore be substantially higher than has been assumed from an analysis of cohorts of subjects with adult T-cell leukaemia or HTLV-1-associated myelopathy. These findings have important implications for epidemiological research and for determining health care priorities in resource-limited settings.

Introduction

The Human T-Lymphotropic Virus type 1 (HTLV-1) is an oncogenic retrovirus that preferentially infects CD4+ T cells[1]. Worldwide, HTLV-1 infects as many as 20 million people who predominantly dwell in areas of high endemicity in south-western Japan and developing countries of the Caribbean basin, South America and sub-Saharan Africa[2]. An endemic focus is present in central Australia[3] where more than 40% of Indigenous adults are HTLV-1cinfected in some remote communities[4].

Clinically significant sequelae of HTLV-1 infection include a haematological malignancy, Adult T cell Leukemia/Lymphoma (ATL), and inflammatory diseases, such as HTLV-1 associated myelopathy/tropical spastic paraparesis (HAM/TSP)[1]. In Japan and the Caribbean, lifetime risks of HAM/TSP and ATL range between 0.3–4% and 1–5%, respectively[1]. Bronchiectasis is the most common clinical manifestation of HTLV-1 infection in Indigenous Australians, amongst whom the adult prevalence of this condition is the highest reported worldwide (>1%)[5,6]. Chest computed tomography has also revealed bronchiectasis in Japanese adults infected with HTLV-1; however, the most frequently reported radiological pattern of HTLV-1 associated pulmonary disease in this population is bronchitis/bronchiolitis[7,8], which has not been described in Indigenous Australians.

In endemic areas in Japan and Africa, HTLV-1 seropositivity is associated with increased mortality[9-12], which has been attributed to non-neoplastic conditions[9,10]. The interpretation of these studies is limited by their inability to control for clinically defined comorbid conditions that might independently increase mortality [9,11,12] [10]. For example, HTLV-1 seropositivity had no effect on mortality in a large hospital-based cohort of Indigenous Australian adults after adjusting for other medical conditions[13]. Given the close association between the number of HTLV-1-infected cells in peripheral blood (the HTLV-1 proviral load, pVL) and serious HTLV-1 associated complications[1,14], any influence of HTLV-1 infection on mortality might be revealed by stratifying outcomes according to HTLV-1 pVL. In central Australia, Indigenous adults with higher HTLV-1c pVL have more extensive, radiologically defined pulmonary injury^[6] and are more likely to present with life-threatening bacterial infections[15]. A single, small study in Guinea-Bissau, where causes of death could not be ascertained, found that mortality increased with HTLV-1 pVL[16]. The present study was therefore commenced to quantify the prognosis of HTLV-1c infection and chronic lung disease and the risk of death according to the HTLV-1c pVL in a hospital-based cohort of Indigenous adults who were well characterized with regard to comorbid conditions and for whom causes of death could be accurately determined in nearly all cases.

Methods

Study setting

Alice Springs Hospital (ASH) is the only medical facility serving central Australia, an area of $>1,000,000 \text{ km}^2$. Critically ill patients are transferred by air to ASH, which has sophisticated diagnostic capabilities.

Recruitment

All Indigenous patients aged >15 years with a discharge diagnosis of bronchiectasis, 1st June 2008 to 31st December 2013, were identified from the ASH patient management database, which coordinates all in-patient and out-patient hospital activities. Indigenous status was determined from self-reported data obtained at admission, as recorded in the patient information database. Potential subjects were offered enrolment when next admitted for >48 hours. Among 165 eligible cases, 154 were recruited (eleven subjects left hospital before recruitment was possible). Written reports for chest high-resolution computed tomography (cHRCT) were reviewed for all subjects, confirming bronchiectasis in 104 cases and bronchitis with or without bronchiolitis in 33 cases (bronchitis alone, 20; bronchitis and bronchiolitis, 12; bronchiolitis alone, 1)(Fig 1). Patients with chronic pulmonary disease were treated according to local guidelines which includes antibiotic therapy for infective exacerbations [17]. A further 686 Indigenous patients aged >15 years who were admitted for >48 hours were prospectively recruited during the same period. These control subjects had no evidence of lower respiratory tract infection at the time of recruitment, no recorded discharge diagnosis of bronchiectasis, and no clinical or radiological evidence of bronchiectasis, Research team members who were unaware of HTLV-1 serostatus were responsible for recruitment (Fig 1).

Demographic and clinical details were extracted from medical records at the time of recruitment using a standardized data-collection form. HTLV-1 associated conditions were identified from medical records at baseline and study end. No control patient developed chronic pulmonary disease during the study period. Mortality data was obtained at study end from the ASH patient management database, and the cause of death was determined from





https://doi.org/10.1371/journal.pntd.0006281.g001

death certificates held in Registries in the Northern Territory of Australia and South Australia. Death certificates were not available for four subjects who died in remote communities in Western Australia, for whom a cause of death was sought from the responsible remote clinic.

Definitions

Bronchitis was diagnosed where cHRCT revealed bronchial wall thickening or dilatation not fulfilling criteria for bronchiectasis, and bronchiolitis where cHRCT revealed multiple

centrilobular nodules or a 'tree-in-bud' pattern[8]. Chronic obstructive pulmonary disease (COPD) required a clinical diagnosis in the medical record and appropriate chest X-ray findings. Emphysema without bronchial wall injury or bronchiolitis was recorded for 18 subjects with COPD examined by cHRCT. Chest HRCT was not performed on 12 subjects who did not meet criteria for such imaging[17]. No subject with symptoms consistent with HAM/TSP received lumbar puncture; the diagnosis was therefore considered 'probable' in all cases. Asymptomatic HTLV-1-infected subjects were those without radiological evidence of airway inflammation or recognized HTLV-1 associated conditions [1]. Residence >80 km from the township of Alice Springs was defined as remote.

Ethics approval

The study was approved by the Central Australian Human Research Ethics Committee. All patients, and their parents/guardians if aged <18 years, gave written informed consent in primary languages.

HTLV-1 serologic and molecular studies

Whole blood samples were collected from each participant at the time of recruitment. Peripheral blood buffy coats (PBBC) were prepared, and plasma and PBBCs were stored at ASH at -80° C until transfer to the National Serology Reference Laboratory, Melbourne. Samples were screened for antibodies to HTLV-1 using both an enzyme immunoassay (Murex HTLV-I + II, DiaSorin, Italy) and a particle agglutination assay (Serodia HTLV-1, Fujirebio, Tokyo, Japan). Any sample reactive on either screening assay was tested by Western blot (HTLV-I/II Blot2.4, MP Biomedicals Asia Pacific Pte. Ltd., Singapore) and HTLV-1c PCR. Primers and fluorescently labelled hydrolysis probes were designed to target a highly conserved 88 bp fragment of the gag gene in the p19 coding region of the Australo-Melanesian HTLV-1 subtype C[18] and multiplexed with primers and probes to the albumin gene[19]. SP cells were used to generate a standard curve from which HTLV-1 pVL (copies per 10⁵ peripheral blood leukocytes; PBL) was calculated. Samples and standards were extracted using the Qiagen QIA blood Mini Extraction kit and the extracts amplified on a Stratagene Mx3000p Real Time PCR Instrument (Integrated Sciences). The extract (5 μ L) was added to 20 μ L of Master mix containing 2 x Brilliant Multiplex QPCR Master Mix (Agilent Technologies) 0.3 µM of each primer (Gene works) and 0.16 µM of each probe (Sigma-Aldrich) and amplified at 95°C for 10 minutes, 45 cycles at 95°C for 30 seconds, 65°C for 60 seconds and 72°C for 60 seconds.

Oligoclonality

The clonality of HTLV-1-infected PBLs was determined by high-throughput sequencing of PBBC cell genomic DNA. The oligoclonality index (OCI) was calculated as previously described[20], and then adjusted to limit underestimation of the OCI due to the small observed number of proviruses[21]. The OCI provides a measure of the non-uniformity of the clone abundance distribution of the infected cell population: OCI = 1 indicates perfect mono-clonality (only one clone constitutes the total proviral load); OCI = 0 indicates perfect polyclonality (all clones have the same abundance)[20]. Samples were selected for clonality analysis if subjects had HTLV-1 pVL >100 copies per 10⁵ PBL, were HBsAg negative and strongyloides seronegative. Although 53 subjects met these criteria, technical difficulties prevented analysis for nine subjects (inadequate number of unique integration sites to accurately determine OCI, 7; unable to sequence integration site, 2). The OCI was therefore compared between 29 asymptomatic and 15 symptomatic subjects (bronchiectasis, 10; bronchitis/bronchiolitis, 3; uveitis, 2).

Statistical analysis

All analysis was performed using Stata version 14.2 (StataCorp, College Station, USA). HTLV-1 pVL was log-transformed and also categorized as low if <1000 and high if ≥1000 per 10⁵ PBL, a cut-off that has been associated with an increased risk of HAM/TSP[22]. Differences between subjects who were HTLV-1 uninfected, those with low HTLV-1 pVL, and those with high HTLV-1 pVL were assessed using ANOVA for continuous variables and chi-squared tests for categorical variables. For statistical purposes, causes of death were grouped into six non-overlapping categories: bronchiectasis, sepsis, cardiovascular disease, malignancy, chronic kidney disease and chronic liver disease. We used survival analysis to determine the association between HTLV-1 pVL and both overall and cause-specific mortality. Subjects were followed until either date of death or 30th March 2015. The association between overall mortality and HTLV-1 pVL was assessed using log-rank tests and Kaplan-Meier curves in univariate analysis and using Cox regression for multivariate analysis. Associations with cause-specific mortality were assessed using competing risks analysis with all causes except the specific cause of interest treated as a competing risk. Where HTLV-1 pVL was treated as a categorical variable we also tested for a trend by creating a continuous variable with value zero for those uninfected, and with the median value of HTLV-1 pVL for those in the low and high pVL categories. Predictors of chronic airways inflammation were assessed using multivariate binary logistic regression. A 2-sided Type 1 error rate of p < 0.05 was regarded as indicating statistical significance in each analysis.

Results

Demographic and clinical characteristics are presented in <u>Table 1</u>. Among 840 subjects recruited to the study, 307 (36.5%) were HTLV-1 infected (HTLV-1 Western blot positive, 268; Western blot indeterminate/HTLV-1c PCR positive, 39).

Clinical associations

Radiologically defined airways inflammation was more common among HTLV-1c-infected subjects (Table 1). Bronchiectasis was confirmed in 59/307 (19.2%) HTLV-1c-infected subjects and 45/533 (8.4%) HTLV-1c uninfected subjects. Similarly, cHRCT revealed bronchitis/bronchiolitis in 21/307 (6.8%) HTLV-1c-infected subjects and 12/533 (2.3%) who were HTLV-1c uninfected (Table 1).

Compared to the median HTLV-1c pVL of asymptomatic subjects (n = 208, 30.4 copies per 10^5 PBL (min, 0.01; max, 18600), those for subjects with bronchiectasis (n = 59, 494 copies per 10^5 PBL; min, 0.01; max, 87900) (p = 0.001) and bronchitis/bronchiolitis (n = 21, 486 copies per 10^5 PBL; min, 0.01; max, 70200)(p = 0.042) were 16-fold higher,(Fig 2). Median HTLV-1c pVL of subjects with any airways inflammation (bronchiectasis and bronchitis/bronchiolitis; 490 copies per 10^5 PBL; min, 0.01; max 87900) was 16-fold higher than that of asymptomatic subjects (p<0.001). Few other recognised causes of bronchiectasis were found (Table 2).

In a multivariate model that controlled for demographic factors, smoking and harmful alcohol consumption, HTLV-1 infection increased the risk of any airways inflammation 2.9-fold (p<0.001) (Table 3), while HTLV-1c pVL \geq 1000 copies per 10⁵ PBL increased the risk 2.2-fold among HTLV-1-infected subjects (p = 0.006) (Table 4).

Other HTLV-1 associated conditions included infective dermatitis (4), crusted scabies (4), probable HAM/TSP (2) and uveitis (2). Four subjects with HTLV-1-associated bronchiectasis had other sequelae of HTLV-1 infection (infective dermatitis, 2; HAM/TSP, 1; uveitis, 1). There was no difference in strongyloides seropositivity between groups (Table 1), nor was there any difference in log-transformed HTLV-1c pVL using linear regression (p = 0.409);

	No HTLV-1 (n = 533)	Low HTLV-1 proviral load (n = 212)	High HTLV-1 proviral load (n = 95)	p-value ¹
Age at test, years (mean±SD)	46.2±15.3	51.6±13.7	51.1±13.8	< 0.001
Male, n (%)	305 (57.2)	119 (56.1)	58 (61.0)	0.72
Adult residency, n (%) ^a				
Urban	75 (14.1)	25 (11.8)	11 (11.6)	0.20
Remote	376 (70.5)	138 (65.1)	64 (67.4)	
Town Camp	81 (15.2)	48 (22.6)	20 (21.1)	
Other States	1 (0.2)	1 (0.5)	0 (0.0)	
Bronchiectasis, n (%) ^b	45 (8.4)	35 (16.5)	24 (25.3)	< 0.001
Bronchitis/Bronchiolitis, n (%) ^c	12 (2.3)	11 (5.2)	10 (10.5)	0.001
COPD, n (%) ^d	16 (3.0)	11 (5.2)	3 (3.2)	0.34
CHF, n (%)	53 (9.9)	22 (10.4)	8 (8.4)	0.91
IHD, n (%)	103 (19.3)	43 (20.3)	16 (16.8)	0.79
Diabetes, n (%)	274 (51.4)	114 (53.8)	51 (53.7)	0.81
ESKD, n (%)	63 (11.9)	30 (14.3)	10 (10.5)	0.60
CLD, n (%)	53 (9.9)	29 (13.7)	12 (12.6)	0.28
Malignancy, n (%)	19 (3.6)	9 (4.2)	5 (5.3)	0.62
Alcohol, n (%) ^e	337 (63.2)	122 (57.6)	48 (50.5)	0.04
Strongyloides serology ^f	111/471 (23.5)	51/180 (28.3)	18/70 (25.7)	0.08
Deaths, n(%)	85 (15.9)	43 (20.2)	27 (28.4)	
Age at death, years (mean±SD)	53.6±13.5	54.8±9.2	51.4±13.4	0.54

Table 1. Demographics and clinical characteristics of indigenous adults according to HTLV-1 status and HTLV-1 proviral load.

a, residence in adulthood. Urban, residence in Alice Springs; Remote, residence in remote community ≥80 km from Alice Springs; Town camp, residence in town camp or community <80 km from Alice Springs; other states, resident outside central Australia

b, chest HRCT proven bronchiectasis

c, chest HRCT findings consistent with bronchitis or bronciolitis

d, clinical and radiological evidence of chronic obstructive pulmonary disease

e, history of harmful alcohol consumption documented in medical records or any hospital admission with alcohol related complications

f, Strongyloides serology was performed for 520, 206 and 88 subjects who were HTLV-1 negative, HTLV-1 infected with low HTLV-1 proviral load and high HTLV-1 proviral load, respectively. Data for 49, 26 and 18 subjects who recorded equivocal serological results were excluded from the analysis.

¹ Difference between the three groups using ANOVA for continuous variables and Fishers Exact test for categorical variables; Abbreviations: COPD, Chronic obstructive pulmonary disease; CHF, Congestive heart failure; IHD, Ischaemic heart disease; ESKD, end-stage kidney disease; CLD, chronic liver disease.

https://doi.org/10.1371/journal.pntd.0006281.t001

median (IQR) HTLV-1c pVL copies per 10⁵ PBL for subjects who were strongyloides seronegative (133; IQR 2.1, 1387), seropositive (61; IQR 1, 1039) and those with equivocal strongyloides serological results (190; IQR 2.7, 2473).

Although the risk of malignancy was not increased among HTLV-1c infected subjects, one subject (HTLV-1c pVL, 5500 copies per 10⁵ PBL) developed ATL during follow-up, one developed penile cancer (HTLV-1c pVL, 70200 copies per 10⁵ PBL) and another metastatic anal cancer (HTLV-1c pVL, 28000 copies per 10⁵ PBL).

Clonality

The median OCI did not differ between asymptomatic (0.401; IQR 0.350, 0.485) and symptomatic groups (0.411; IQR 0.355, 0.552) (p = 0.51) or when symptomatic subjects with uveitis were excluded from the analysis (symptomatic group median OCI 0.429; IQR 0.369, 0.552) (p = 0.341). Median \log_{10} HTLV-1c pVL copies per 10⁵ PBL in subjects selected for clonality analysis did not differ between asymptomatic subjects (0.947; IQR 0.342, 1.964) and those with airways inflammation (1.50; IQR 0.669, 4.074)(p = 0.20).



Fig 2. HTLV-1c proviral load compared between asymptomatic subjects, bronchiolitis/bronchitis and bronchiectasis. Subjects with chronic airways disease were examined by chest high resolution computed tomography. Median (IQR) HTLV-1c pVL for subjects with bronchiolitis/bronchitis (2.69 (0.82, 3.43) \log_{10} copies per 10⁵ PBL) and bronchiectasis (2.69 (1.57, 3.49) \log_{10} copies per 10⁵ PBL) were significantly higher than that of the asymptomatic group (1.48 (-0.001, 2.97) \log_{10} copies per 10⁵ PBL)(asymptomatic vs bronchiectasis, p = 0.001; asymptomatic vs bronchitis/bronchiolitis, p = 0.042). Asymptomatic subjects exclude: i) those with other HTLV-1 associated conditions (infective dermatitis, 2; probable HAM/ TSP, 2; uveitis, 2; crusted scabies, 4; ATL; 1), ii) five subjects with a discharge diagnosis of bronchiectasis without radiological evidence of bronchiectasis, bronchitis or bronchiolitis and iii) three subjects who presented with neurological symptoms for whom HAM/TSP could not be excluded due to cognitive impairment.

https://doi.org/10.1371/journal.pntd.0006281.g002

Mortality

During 2140 person-years of follow-up, 155 deaths were recorded (HTLV-1 uninfected, 85; HTLV-1 infected, 70). Non-bronchiectasis causes of death were infections (37)(lower

Table 2.	Investigations	for causes of radio	logically	v defined airwa	vs inflammation	n according to HT	LV-1 serostatus.
			· · · · · · ·			_	

	AFB	ANA	ASP	Ig Deficiency	IgG subclass Deficiency	A1AT Deficiency
HTLV-1 Uninfected	0/39	2/48 ^a	0/47	1/48 ^b	1/44 ^b	0/42
HTLV-1 Infected	1/60 ^c	1/59 ^d	0/53	0/53	0/49	0/50

Denominator gives number of subjects who received each test.

a, bronchitis/bronchiolitis associated with Sjogren's syndrome (1), rheumatoid arthritis (1)

b, bronchiectasis associated with Immunoglobulin A and $\mathrm{G}_{\mathrm{1-3}}$ deficiency

c, *Mycobacterium avium/intracellulare* complex repeatedly cultured from sputum in one case with bronchitis/bronchiolitis (HTLV-1c pVL, 9.4 copies per 10⁵ PBL) d, Antinuclear antibody positive but not deemed to be significant for a 32 year old, HTLV-1c infected man who died with rapidly progressive haemophagocytic lymphohistiocytosis (HTLV-1c pVL, 1951 copies per 10⁵ PBL). Abbreviations: AFB, \geq 3 adequate sputum cultures for mycobacteria or any case in which a mycobacterium was isolated; ANA, anti-nuclear antibodies; A1AT deficiency: deficiency of alpha 1 anti-trypsin; ASP, aspergillus precipitins; Ig, immunoblobulin; Ig Deficiency, levels of Immunoglobulin classes A or G less than lower limit of normal; IgG subclass deficiency, levels of IgG subclasses less than lower limit of normal.

https://doi.org/10.1371/journal.pntd.0006281.t002

Table 3. Multivariable predictors of radiologically defined airway inflammation^a among all subjects¹ (n = 840)^b.

	Model 1		Model 2		Model 3	
	OR (95% CI)	p-value ¹	OR (95% CI)	p-value ¹	OR (95% CI)	p-value ¹
HTLV-1 Infected	2.9 (2.0, 4.3)	< 0.001	2.8 (1.9, 4.2)	< 0.001	2.9 (2.0, 4.3)	< 0.001
Age ^c			1.0 (0.99, 1.02)	0.28	1.0 (0.99, 1.02)	0.29
Gender			1.11 (0.8, 1.6)	0.60	1.2 (0.78, 1.80)	0.43
Residence ^d Urban					1.00	-
Remote					0.8 (0.4, 1.4)	0.39
Town Camp Other States					1.1 (0.6, 2.1) 3.7 (0.2, 64.7)	0.77 0.38
Ever smoked					0.8 (0.5, 1.2)	0.30
Alcohol ^e					1.5 (0.9, 2.2)	0.09

a, bronchiectasis, bronchitis or bronchiolitis identified by chest high resolution computed tomography

b, HTLV-1 seronegative, 533; HTLV-1 infected, 307 (Western blot positive, 268; HTLV-1 Western blot indeterminate/HTLV-1 PCR positive, 39)

c, per 1 year increase in age

d, Residence in adulthood

e, history of harmful alcohol consumption documented in medical records or any hospital admission with alcohol related complications. Urban, residence in Alice Springs; Remote, residence in remote community \geq 80 km from Alice Springs; Town camp, residence in town camp or community <80 km from Alice Springs; other states, resident outside central Australia.

¹ Using multivariate binary logistic regression. Model 1: unadjusted, Model 2: adjusted for age and gender, Model 3: additionally adjusted for area of residence, previous smoking history and alcohol abuse. p-value based on the Wald test of the associated odds ratio beta coefficient.

https://doi.org/10.1371/journal.pntd.0006281.t003

respiratory tract infections, 15), cardiovascular disease (37), malignancy (15), end-stage kidney disease (ESKD) (15), chronic liver disease (10), intracerebral haemorrhage (4), primary pulmonary hypertension (2) and amyloidosis (1).

Table 4. Multivariable predictors of bronchiectasis^a among HTLV-1 infected subjects¹ (n = 307).

-	Model 1		Model 2		Model 3	
	OR (95% CI)	p-value	OR (95% CI)	p-value	OR (95% CI)	p-value
pVL Low ^b pVL High ^c	1.00 2.0 (1.2, 3.4)	0.01	1.00 2.0 (1.2, 3.4)	0.01	1.0 2.2 (1.2, 3.7)	0.006
Age ^d			1.0 (0.98, 1.01)	0.61	1.0 (0.98, 1.01)	0.63
Gender			1.4 (0.8, 2.3)	0.25	1.4 (0.8, 2.6)	0.22
Residence ^e Urban					1.00	-
Remote					1.1 (0.5, 2.6)	0.84
Town Camp					1.7 (0.6, 4.4)	0.29
Ever smoked					0.8 (0.4, 1.5)	0.47
Alcohol ^f					1.6 (0.9, 2.9)	0.10

a, diagnosed by chest high resolution computed tomography

b, pVL low, HTLV-1 proviral load<1000 copies per 10⁵ PBL

c, pVL high, HTLV-1c proviral load ≥ 1000 copies per 10^5 PBL

d, per 1 year increase in age

e, Residence in adulthood

f, history of harmful alcohol consumption documented in medical records or any hospital admission with alcohol related complications. Urban, residence in Alice Springs; Remote, residence in remote community \geq 80 km from Alice Springs; Town camp, residence in town camp or community <80 km from Alice Springs; other states, resident outside central Australia.

¹ Using multivariate binary logistic regression. Model 1: unadjusted, Model 2: adjusted for age and gender, Model 3: additionally adjusted for area of residence, previous smoking history and alcohol abuse. p-value based on the Wald test of the associated odds ratio beta coefficient.

https://doi.org/10.1371/journal.pntd.0006281.t004

Subjects with high HTLV-1c pVL were more likely to die during the study period (Fig 3). Mortality rates for high HTLV-1c pVL, low HTLV-1c pVL and HTLV-1 uninfected subjects were 28.4% (27/95), 20.2% (43/212) and 15.9% (85/533), respectively (p = 0.011). The unadjusted HR for death among subjects with low and high HTLV-1c pVL were 1.24 (95% CI, 0.85, 1.81) and 1.75 (95% CI, 1.13, 2.670), respectively (p = 0.021 for trend). The statistical significance was diminished after adjusting for age, gender, place of residence and harmful alcohol consumption (p = 0.084). The effect of HTLV-1c pVL on overall mortality was lost in a multivariate model that included bronchiectasis (aHR, 1.045; 95% CI, 0.658–1.660) (Table 5). Other predictors of death were age at test, male gender and comorbid conditions (Table 5).

Cause-specific mortality. Relative to HTLV-1c uninfected subjects, unadjusted hazard ratios for a bronchiectasis-related death were 4.73 (95% CI, 2.05, 10.9) and 1.44 (95% CI, 1.02, 6.49) for subjects with high and low HTLV-1c pVL, respectively (Table 6).

The effect of HTLV-1c pVL on bronchiectasis-related mortality remained significant for subjects with high HTLV-1c pVL (4.31; 95% CI, 1.78, 10.4) when other causes of death were treated as competing risks and after adjusting for alcohol, place of residence, age at test, and gender (Table 6) (Fig 4). The risk of a bronchiectasis-related death increased 4.5% with each 100 HTLV-1c copies per 10^5 PBL when median values for proviral load were treated as a continuous variable in an adjusted model (p = 0.009 for trend). This effect was not apparent among the 736 subjects without bronchiectasis (Table 7). Causes of bronchiectasis-related deaths included lower respiratory tract infections (15), respiratory failure (13) and haemoptysis (1).





https://doi.org/10.1371/journal.pntd.0006281.g003

	HR	95% CI	p-value ¹
Age at test ^a	1.025	1.011-1.038	< 0.001
Male gender	1.848	1.275-2.677	0.001
Adult residence ^b ,	reference	0.492-1.407	0.493
Urban	0.832	0.810-2.534	0.217
Remote	1.432	0.158-10.438	0.815
Town Camp	1.286		
Other			
No HTLV-1	1.000	-	-
HTLV-1 pVL low ^c	0.843	0.571-1.245	0.391
HTLV-1 pVL high ^d	1.040	0.657-1.650	0.865
Bronchiectasis ^e	3.534	2.293-5.445	< 0.001
COPD ^f	0.761	0.476-1.218	0.256
CHF	3.289	2.089-5.178	< 0.000
IHD	1.694	1.150-2.494	0.008
Diabetes	0.972	0.658-1.434	0.885
ESKD	2.049	1.317-3.190	0.001
CLD	2.608	1.792-3.793	< 0.001
Malignancy	4.101	2.472-6.803	< 0.001
Alcohol ^g	0.702	0.488-1.008	0.055

Table 5. Multivariate predictors of death among 840 Indigenous adults.

a, per 1 year increase in age at test

b, Residence in a dulthood. Urban, residence in Alice Springs; Remote, residence in a community $\geq \! 80$ km from Alice Springs; Town Camp, residence in town camp or community $<\! 80$ km from Alice Springs

c, HTLV-1 proviral load <1000 copies per 10⁵ peripheral blood leukocytes

d, HTLV-1 proviral load \geq 1000 copies per 10⁵ peripheral blood leukocytes

e, chest HRCT proven bronchiectasis

f, clinical and radiological evidence of chronic obstructive pulmonary disease

g, history of harmful alcohol consumption documented in medical records or any hospital admission with alcohol related complications. Abbreviations: COPD, Chronic obstructive pulmonary disease; CHF, Congestive heart failure; IHD, Ischaemic heart disease; ESKD, end-stage kidney disease; CLD, chronic liver disease.

¹p-value based on the Wald test of the associated hazard ratio beta coefficient.

https://doi.org/10.1371/journal.pntd.0006281.t005

Mean (±SD) age at death for bronchiectasis-related deaths (49.5±15.3 years) was lower than that of subjects who died due to cardiovascular disease (55.0±11.5 years)(p = 0.073), sepsis (56.7±16.7)(p = 0.044) or malignancy (60.1±9.3) (p = 0.007).

Discussion

In a large hospital-based cohort of Indigenous Australian adults, a higher baseline HTLV-1c pVL prospectively predicted a bronchiectasis-related death, which occurred at a mean age of only 49.5 years. In addition to confirming a previously reported association between HTLV-1c infection and bronchiectasis[5], the present study also revealed an association with bronchitis/ bronchiolitis. Airways inflammation was strongly associated with higher HTLV-1c pVL[6,15]. The median HTLV-1c pVL of subjects with radiologically defined airways inflammation was 16-fold higher than that for asymptomatic HTLV-1-infected subjects, and risk of airway inflammation increased three-fold among subjects with higher HTLV-1c pVL in an adjusted model.

HTLV-1 associated inflammatory diseases are thought to result from a genetically determined, inefficient cytotoxic T lymphocyte response, permitting widespread dissemination of the virus in a large number of HTLV-1-infected T-cell clones, which is reflected in a high

		HTLV-1 Negative N = 533 (63.5%)	HTLV-1 pVL Low N = 212 (25.2%)	HTLV-1 pVL High N = 95 (11.3%)	P-value for trend ¹
All-cause mortality	Events, n (%)	83 (55.0)	41 (27.1)	27 (17.9)	
	Unadjusted	1.00	1.24 (0.85, 1.81)	1.75 (1.13, 2.70)	0.02
	Multivariate ²	1.00	1.00 (0.69, 1.47)	1.43 (0.92, 2.23)	0.09
Cause-specific mortality ³					
Bronchiectasis	Events, n (%)	12 (41.4)	7 (24.1)	10 (34.5)	
	Unadjusted	1.00	1.44 (1.02, 6.49)	4.73 (2.05, 10.94)	<0.001
	Multivariate ²	1.00	1.23 (0.69, 4.58)	4.31 (1.78, 10.42)	0.001
CVD	Events, n (%)	25 (67.6)	8 (21.6)	4 (10.8)	
	Unadjusted	1.00	0.78 (0.35, 1.74)	0.81 (0.29, 2.30)	0.78
	Multivariate ²	1.00	0.66 (0.30, 1.46)	0.66 (0.24, 1.84)	0.57
Liver Disease	Events, n (%)	4 (40.0)	5 (50.0)	1 (10.0)	
	Unadjusted	1.00	3.10 (0.83, 11.6)	1.38 (0.16, 11.6)	0.89
	Multivariate ²	1.00	2.84 (0.75, 10.8)	1.30 (0.16, 10.8)	0.81
Malignancy	Events, n (%)	9 (60.0)	4 (26.7)	2 (13.3)	
	Unadjusted	1.00	1.08 (0.34, 3.43)	1.05 (0.23, 4.70)	0.98
	Multivariate ²	1.00	0.74 (0.20, 2.83)	0.74 (0.16, 3.44)	0.79
Other	Events, n (%)	6 (60.0)	3 (30.0)	1 (10.0)	
	Unadjusted	1.00	1.23 (0.31, 4.92)	0.77 (0.10, 5.95)	0.75
	Multivariate ²	1.00	1.13 (0.27, 4.7)	0.71 (0.09, 5.41)	0.71
Renal Disease	Events, n (%)	11 (68.8)	4 (25.0)	1 (6.2)	
	Unadjusted	1.00	0.88 (0.28, 2.75)	0.45 (0.06, 3.56)	0.46
	Multivariate ²	1.00	0.74 (0.24, 2.30)	0.39 (0.05, 3.04)	0.42
Sepsis	Events, n (%)	19 (51.3)	10 (27.0)	8 (21.6)	
	Unadjusted	1.00	1.26 (0.58, 2.70)	2.17 (0.95, 4.96)	0.08
	Multivariate ²	1.00	1.07 (0.50, 2.30)	1.81 (0.77, 4.29)	0.16

Table 6. Hazard ratios for all-cause and cause-specific mortality according to HTLV-1 proviral load status.

Hazard ratios were assessed using Cox-regression for all-cause mortality and competing risks regression for cause-specific mortality.

¹ p-value for test of HTLV-1 pVL trend across the 3 categories conducted using the median pVL value for each category.

²Adjusting for alcohol, place of adult residence (Alice Springs, Remote, Town camp or other States), age at test and gender.

³All other causes of death were treated as a competing risk. HTLV-1c pVL low, proviral load <1000 copies per 10⁵ peripheral blood leukocytes; HTLV-1c pVL high, proviral load ≥ 1000 copies per 10⁵ peripheral blood leukocytes. Abbreviations: CVD, cardiovascular disease. Cause of death not determined for four subjects

https://doi.org/10.1371/journal.pntd.0006281.t006

HTLV-1 pVL[23]. Organ infiltration by HTLV-1-infected lymphocytes then leads to high local HTLV-1 antigen levels, provoking an immune response and tissue injury following the release of pro-inflammatory cytokines and chemokines[23]. Although this has been best studied for the prototypical HTLV-1 associated disease, HAM/TSP, HTLV-1 infection is also associated with inflammation in other organs[14] including the lungs [24]. Consistent with the presumed mechanism of pathogenesis of HAM/TSP [23], pulmonary involvement is associated with infiltration of HTLV-1-infected lymphocytes[25,26], increased *tax/rex* mRNA[27] expression, and an inflammatory cytokine milieu in bronchoalveolar lavage fluid[27]. In large Japanese case series, cHRCT was abnormal in 30–61% of HTLV-1 infected subjects of whom 23.6–29.5% had a bronchitis/bronchiolitis pattern of disease and 15.6–22.5% had frank bronchiectasis[7,8]. The pathological correlate of these observations is lymphocyte infiltration in bronchiole walls[7]. Persistent HTLV-1-mediated airways inflammation may therefore lead to progressive bronchial wall dilatation, and bronchiectasis. High rates of bronchiectasis among HTLV-1-infected Japanese adults[7,8], and associations between HAM/TSP and



Fig 4. Cumulative death rate due to bronchiectasis. Estimated using competing-risks Cox regression among 533 HTLV-1 uninfected subjects, 212 subjects with low HTLV-1 proviral load (<1000 copies/10⁵ peripheral blood leukocytes) and 95 with high HTLV-1c proviral load (\geq 1000 copies/10⁵ peripheral blood leukocytes).

https://doi.org/10.1371/journal.pntd.0006281.g004

bronchiectasis in UK [28] and Brazilian cohorts [29] suggest that HTLV-1-associated bronchiectasis affects individuals of diverse genetic backgrounds infected with HTLV-1 strains other than HTLV-1c. Although HTLV-1 associated pulmonary disease in Japan is thought to be largely sub-clinical[14], published clinical details are limited and prospective survival studies have not been performed.

Table 7. Effect of HTLV-1c pVL on all-cause mortality according to bronchiectasis status.

		HTLV-1 Negative	HTLV-1 pVL Low	HTLV-1 pVL High	P-value for trend ¹
No Bronchiectasis		N = 488 (66.3%)	N = 177 (24.0%)	N = 71 (9.7%)	
	Events, n (%)	69 (59.5)	34 (29.3)	13 (11.2)	
	Unadjusted	1.00	1.34 (0.89, 2.02)	1.21 (0.67, 2.19)	0.72
	Multivariate ²	1.00	1.10 (0.72, 1.67)	0.89 (0.49, 1.63)	0.63
Bronchiectasis		N = 45 (43.2%)	N = 35 (33.7%)	N = 24 (23.1%)	
	Events, n (%)	14 (40.0)	7 (20.0)	14 (40.0)	
	Unadjusted	1.00	0.62 (0.25, 1.54)	1.75 (0.83, 3.72)	0.03
	Multivariate ²	1.00	0.50 (0.20, 1.29)	2.04 (0.89, 4.69)	0.009

Hazard ratios were assessed using Cox-regression for all-cause mortality and competing risks regression for cause-specific mortality.

¹p-value for test of HTLV-1 pVL trend across the 3 categories conducted using the median pVL value for each category.

²Adjusting for alcohol, place of adult residence (Alice Springs, Remote, Town camp or other States), age at test and gender. All other causes of death were treated as a competing risk.

https://doi.org/10.1371/journal.pntd.0006281.t007

Consistent with other HTLV-1 associated inflammatory diseases[1,14], the median baseline HTLV-1c pVL of subjects with airways inflammation was substantially higher than that of asymptomatic subjects. For example, the median HTLV-1c pVL in subjects with HAM/TSP is between 7-fold and 16-fold greater than that of asymptomatic subjects[22][28][30–32]. Among subjects with HAM/TSP, higher HTLV-1 pVL correlates with more severe motor weakness [31] and more rapid neurological progression[32]. We previously demonstrated that HTLV-1-infected Indigenous adults have more diffuse bronchiectasis[5] and that a higher HTLV-1c pVL correlates with more extensive pulmonary injury[6]. In contrast to HTLV-1-mediated inflammation in other tissues, pulmonary parenchymal injury can result in directly life-threat-ening complications, including respiratory failure[5]. Among subjects with airways disease in whom HTLV-1 oligoclonality could be studied, there was no difference in the median OCI when compared to that of asymptomatic subjects. This suggests that higher HTLV-1c pVL were due to an increased number of infected clones rather than clonal expansion, which is consistent with the conclusion previously reported for subjects with HAM/TSP[20]

Increased mortality due to a specific HTLV-1-associated inflammatory disease has not been prospectively demonstrated previously. However, an excess mortality that is not attributable to ATL or currently recognized HTLV-1-associated inflammatory diseases has been reported in other endemic areas. Adjusted hazard ratios of death are 1.3[10] to 1.77–1.87[9] in Japanese outpatient cohorts, and 3.8 and 2.3 for young and middle-aged adults, respectively, in a community-based cohort in Guinea-Bissau[11,12]. In a study that included only 48 HTLV-1-infected subjects in Guinea-Bissau, mortality was associated with higher HTLV-1 pVL[16]. In Japan, excess mortality was attributed to non-neoplastic diseases, most commonly unspecified kidney and cardiac conditions[9,10]. Although the ASH cohort included subjects with established ESKD and heart disease, HTLV-1 infection was only associated with bronchiectasis-related deaths, and this effect was only revealed after stratifying by HTLV-1c pVL. The difference between studies in the clinical conditions associated with excess mortality may reflect the high burden of illness in our hospital-based cohort, the inability to control for comorbid conditions in other studies, and differences in the social circumstances of the various study populations.

The strengths of this prospective cohort study include the recruitment of nearly all eligible subjects with a discharge diagnosis of bronchiectasis, the use of cHRCT for diagnosis and the blinding of ASH researchers to the HTLV-1 serostatus of subjects and of those who performed HTLV-1 studies to their clinical state. Nevertheless, some design limitations must be recognized. First, investigations to exclude other causes of airways disease could not be performed in all cases. However, consistent with previous studies [5,6], a specific aetiology was rarely found among >70% of subjects who were screened for conditions generally associated with bronchiectasis[17]. Although an effect of childhood respiratory infections cannot be excluded in the present study, we previously found HTLV-1 infection to be the major predictor of adult bronchiectasis in a case-control study that controlled for such infections[6]. Second, only subjects with a discharge diagnosis of bronchiectasis were specifically targeted for recruitment. Twelve subjects with COPD (HTLV-1 infected, 4; HTLV-1 uninfected, 8) were incidentally recruited but not examined by cHRCT because they did not clinically warrant further imaging [17]. The contribution of HTLV-1c infection to less severe respiratory disease than that associated with a discharge diagnosis of bronchiectasis, and the validity of our conclusions in a community setting, require further study. Finally, the absence of an association between strongyloides seropositivity and higher HTLV-1c pVL may be due to the fact that strongyloides serology was assayed in subjects without symptomatic strongyloidiasis.

In summary, HTLV-1c infection and higher HTLV-1c pVL were strongly linked to airways inflammation in a hospital-based cohort of Indigenous Australian adults. Furthermore, higher

baseline HTLV-1c pVL prospectively predicted death due to bronchiectasis, which may result from more extensive disease[5,6], predisposing to life-threatening complications[5] among subjects who are unable to control HTLV-1 replication[6]. Elucidating the causes of higher mortality among people infected with HTLV-1 is relevant to an estimated 20 million people living with HTLV-1 infection in resource-poor areas[2] where the impact of HTLV-1 infection has been little studied.

Supporting information

S1 Checklist. Strobe checklist. (DOC)

Acknowledgments

We acknowledge the assistance of Magdalene Lynch, Lena McCormack and Clinton Pepperill who contributed to patient recruitment and the administrative assistance of Maria Flynn and Sheela Joseph, and gratefully acknowledge the support of the Pathology Department at Alice Springs Hospital. The authors also thank Aviva Witkover and Anat Melamed for help with the linker-mediated PCR clonality analysis.

Author Contributions

Conceptualization: Lloyd Einsiedel.

Data curation: Lloyd Einsiedel, Hai Pham, Rebecca Walley.

Formal analysis: Lloyd Einsiedel, Jocelyn Turpin, Charles Bangham, Richard J. Woodman.

Funding acquisition: Lloyd Einsiedel.

Investigation: Lloyd Einsiedel, Kim Wilson, Rebecca Walley, Jocelyn Turpin, Charles Bangham.

Methodology: Lloyd Einsiedel, Antoine Gessain.

Project administration: Lloyd Einsiedel, Hai Pham.

Software: Hai Pham.

Writing - original draft: Lloyd Einsiedel.

Writing – review & editing: Lloyd Einsiedel, Jocelyn Turpin, Charles Bangham, Antoine Gessain, Richard J. Woodman.

References

- Verdonck K, Gonzalez E, Van Dooren S, Vandamme A, Vanham G, et al. (2007) Human T-lymphotropic virus 1: recent knowledge about an ancient infection. Lancet Infect Dis 7: 266–281. <u>https://doi.org/10. 1016/S1473-3099(07)70081-6 PMID: 17376384</u>
- Gessain A, Cassar O (2012) Epidemiological aspects and world distribution of HTLV-1 infection. Front Microbiol 3: 388. https://doi.org/10.3389/fmicb.2012.00388 PMID: 23162541
- Cassar O, Einsiedel L, Afonso PV, A G (2013) HTLV type 1 subtype C molecular variants among Indigenous Australians: New insights on HTLV-1 molecular epidemiology in Australo-Melanesia PLOS Negl Trop Dis 7: e2418. https://doi.org/10.1371/journal.pntd.0002418 PMID: 24086779
- Einsiedel L, Pham H, Woodman R, Pepperill C, Taylor K (2016) The prevalence and clinical associations of HTLV-1 infection in a remote Indigenous community. Med J Aust 205: 305–309. PMID: 27681971

- Einsiedel L, Fernandes L, Spelman T, Steinffort D, Gotuzzo E (2012) Bronchiectasis is associated with Human T-Lymphotropic Virus 1 infection in an Indigenous Australian population. Clin Infect Dis 54: 43– 50. https://doi.org/10.1093/cid/cir766 PMID: 22095566
- Einsiedel L, Cassar O, Goeman E, Spelman T, Au V, et al. (2014) Higher Human T-Lymphotropic Virus Type 1 Subtype C Proviral Loads Are Associated With Bronchiectasis in Indigenous Australians: Results of a Case-Control Study. Open Forum Infectious Diseases, Volume 1, Issue 1, 1 March 2014, ofu023, https://doi.org/10.1093/ofid/ofu023
- Okada F, Ando Y, Yoshitake S, Yotsumoto S, Matsumoto S, et al. (2006) Pulmonary CT findings in 320 carriers of human T-lymphotropic virus type 1. Radiology 240: 559–564. <u>https://doi.org/10.1148/radiol.</u> 2402050886 PMID: 16864677
- Yamashiro T, Kamiya H, Miyara T, Gibo S, Ogawa K, et al. (2012) CT scans of the chest in carriers of human T-cell lymphotropic virus type 1: presence of interstitial pneumonia. Academic Radiology 19: 952–957. https://doi.org/10.1016/j.acra.2012.03.020 PMID: 22578413
- 9. Iwata K, Ito S, Saito H, Ito M, Nagatomo M, et al. (1994) Mortality among inhabitants of an HTLV-I endemic area in Japan. Jpn J Cancer Res 85: 231–237. PMID: 8188520
- Arisawa K, Sobue T, Yoshimi I, Soda M, Shirahama S, et al. (2003) Human T-lymphotropic virus type-1 infection, survival and cancer risk in southwestern Japan: a prospective study. Cancer Causes and Control 14: 889–896. PMID: 14682446
- Holmgren B, da Silva Z, Vastrup P, Larsen O, Andersson S, et al. (2007) Mortality associated with HIV-1, HIV-2, and HTLV-I single and dual infections in a middle-aged and older population in Guinea-Bissau. Retrovirology 4: 85. https://doi.org/10.1186/1742-4690-4-85 PMID: 18042276
- 12. van Tienen C, Schim van der Loeff M, Peterson I, Cotten M, Andersson S, et al. (2011) HTLV-1 and HIV-2 infection are associated with increased mortality in a rural West African community. PLoS ONE 6: e29026. https://doi.org/10.1371/journal.pone.0029026 PMID: 22194980
- Einsiedel L, Spelman T, Goeman E, Cassar O, Arundell M, et al. (2014) Clinical Associations of Human T-Lymphotropic Virus Type 1 Infection in an Indigenous Australian Population. PLoS Negl Trop Dis 8: e2643 https://doi.org/10.1371/journal.pntd.0002643 PMID: 24454973
- Martin F, Taylor G, Jacobson S (2014) Inflammatory manifestations of HTLV-1 and their therapeutic options. Expert Rev Clin Immunol 10: 1531–1546. <u>https://doi.org/10.1586/1744666X.2014.966690</u> PMID: 25340428
- Einsiedel L, Cassar O, Spelman T, Joseph S, Gessain A (2016) Higher HTLV-1c proviral loads are associated with blood streaminfections in an Indigenous Australian population. J Clin Virol 78: 93–98. https://doi.org/10.1016/j.jcv.2016.03.006 PMID: 27011343
- Ariyoshi K, Berry N, Cham F, Jaffar S, Schim van der Loeff M, et al. (2003) Quantification of Human Tlymphotropic virus type I (HTLV-I) provirus load in a rural West African population: no enhancement of human immunodeficiency virus type 2 pathogenesis, but HTLV-I provirus load relates to mortality. J Infect Dis 188: 1648–1651. https://doi.org/10.1086/379780 PMID: 14639534
- Chang AB, Bell SC, Torzillo PJ, King PT, Maguire GP, et al. (2015) Thoracic Society of Australia and New Zealand guidelines. Chronic suppurative lung disease and bronchiectasis in children and adults in Australia and New Zealand. Med J Aust 202: 21–24. PMID: 25588439
- Einsiedel L, Woodman R, Flynn M, Wilson K, Cassar O, et al. (2016) Human T-Lymphotropic Virus type 1 infection in an Indigenous Australian population: epidemiological insights from a hospital-based cohort study. BMC Public Health 16: 787. https://doi.org/10.1186/s12889-016-3366-5 PMID: 27526923
- Desire N, Dehee A, Schneider V, Jacomet C, Goujon C, et al. (2001) Quantification of Human Immunodeficiency Virus Type 1 Proviral Load by a TaqMan Real-Time PCR Assay. J Clin Microbiol 39: 1303– 1310. https://doi.org/10.1128/JCM.39.4.1303-1310.2001 PMID: 11283046
- Gillet N, Cook L, Laydon D, Hlela C, Verdonck K, et al. (2013) Strongyloidiasis and Infective Dermatitis Alter Human T Lymphotropic Virus-1 Clonality in vivo. PLoS Pathog 9: e1003263. <u>https://doi.org/10.1371/journal.ppat.1003263</u> PMID: 23592987
- 21. Majumder A, Kusago T (2013) A discreet approach to study the distribution-free downward biases of Gini coefficient and the methods of correction in cases of small observations Working Paper Series, ECINEQ 2013: 298.
- 22. Nagai M, Usuku K, Matsumoto W, Kodama D, Takenouchi N, et al. (1998) Analysis of HTLV-1 proviral load in 202 HAM/TSP patients and 243 asymptomatic HTLV-1 carriers: high proviral load strongly predisposes to HAM/TSP. J Neurovirol 4: 586–593. PMID: 10065900
- Saito M, Bangham C (2012) Immunopathogenesis of Human T-Cell Leukemia Virus Type-1-Associated Myelopathy/Tropical Spastic Paraparesis: Recent Perspectives. Leukemia Research and Treatment 2012: 12.
- 24. Sugimoto M, Nakashima H., Watanabe S., Uyama E., Tanaka F., Ando M., Araki S., Kawasaki S. (1987) T-lymphocyte alveolitis in HTLV-1 associated myelopathy. Lancet ii: 1220.

- 25. Mori S, Mizoguchi A, Kawabata M, Fukunaga H, Usuku K, et al. (2005) Bronchoalveolar lavage lymphocytes correlate with HTLV-1 proviral load in HTLV-1 carriers. Thorax 60: 138–143. <u>https://doi.org/10.</u> 1136/thx.2004.021667 PMID: 15681503
- Kawabata T, Higashimoto I, Takashima H, Izumo S, Kubota R (2012) Human T-lymphotropic virus type I (HTLV-I)-specific CD8+ cells accumulate in the lungs of patients infected with HTLV-I with pulmonary involvement. Journal of Medical Virology 84: 1120–1127. <u>https://doi.org/10.1002/jmv.23307</u> PMID: 22585731
- 27. Yamazato Y, Miyazato A, Kawakami K, Satomi Y, Kaneshima H, et al. (2003) High Expression of p40tax and proinflammatory cytokines and chemokines in the lungs of Human T-Lymphotropic Virus Type-1-related bronchopulmonary disorder. Chest 124: 2283–2292. PMID: 14665512
- **28.** Honarbakhsh S, Taylor G (2015) High prevalence of bronchiectasis is linked to HTLV-1-associated inflammatory disease. BMC Infect Dis 5: 258.
- 29. Magno Falcão LF FA, Medeiros Sousa RC, et al. (2017) CT Chest and pulmonary functional changes in patients with HTLV-associated myelopathy in the Eastern Brazilian Amazon. PLoS ONE 12: e0186055. https://doi.org/10.1371/journal.pone.0186055 PMID: 29095831
- 30. Demontis M, Hilburn S, Taylor G (2013) Human T cell lymphotropic virus type 1 viral load variability and long-term trends in asymptomatic carriers and in patients with human T cell lymphotropic virus type 1-related diseases. AIDS Res Hum Retroviruses 29: 359–364. https://doi.org/10.1089/AID.2012.0132 PMID: 22894552
- 31. dos Santos Brito Silva Furtado M, Andrade R, Romanelli L, Ribeiro M, Ribas J, et al. (2012) Monitoring the HTLV-1 Proviral Load in the Peripheral Blood of Asymptomatic Carriers and Patients With HTLV-Associated Myelopathy/ Tropical Spastic Paraparesis From a Brazilian Cohort: ROC Curve Analysis to Establish the Threshold for Risk Disease. J Med Virol 84: 664–671. https://doi.org/10.1002/jmv.23227 PMID: 22337307
- Olindo S, Lezin A, Cabre P, Merle H, Saint-Vil M, et al. (2005) HTLV-1 proviral load in peripheral blood mononuclear cells quantified in 100 HAM/TSP patients: A marker of disease progression. J Neurol Sci 237 53–59. https://doi.org/10.1016/j.jns.2005.05.010 PMID: 15972218